Author's response to reviews

Title: False rumours of disease outbreaks caused by infectious myonecrosis virus (IMNV) in the whiteleg shrimp in Asia

Authors:

Saengchan Senapin (saengchan@biotec.or.th)
Kornsunee Phiwsaiya (kornsunee.phi@biotec.or.th)
Warachin Gangnonngiw (warachin101@hotmail.com)
Timothy W Flegel (sctwf@mahidol.ac.th)

Version: 3 Date: 31 July 2011

Author's response to reviews: see over
18 May 2011

Editors
BMC Journal of Negative Results in Biomedicine

Dear Sirs:

RE MS: False rumours of disease outbreaks caused by infectious myonecrosis virus (IMNV) in the whiteleg shrimp in Asia
By Saengchan Senapin, Kornsunee Phiw saiya, Warachin G angnonngiw and Timothy W. Flegel

We have revised the manuscript making the format changes requested and altering typographical errors. The changes are outlined below.

Yours truly,

Prof. T.W. Flegel
Centex Head

Requested Formatting Changes:

Major revisions (we require the author to make these changes)

Author Contributions: Please confirm that all authors read and approved the final manuscript.

RESPONSE: The sentence has been added.

Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9. Currently figure 2 is cited first.

RESPONSE: Reference to Fig. 1 now comes first on Page 5 followed by reference to Fig. 2.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line Numbers: Please remove all forms of line numbers present in the manuscript.

RESPONSE: The line numbers have been removed.

Tracking: Please remove all forms of tracking and coloured text present in the manuscript.

RESPONSE: Tracking and colored text have been removed.

Tables: Please remove any visible vertical lines present in any tables in the manuscript. Also please note that we are unable to display any merged cells that go across rows. Please reformat your tables with these elements in mind.

RESPONSE: Vertical lines have been removed and there are now no merged cells across rows.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

RESPONSE: We have checked the MS for typographical errors and corrected what we found.